Author’s response to reviews

Title: MUAC-for-age more useful than absolute MUAC for nutritional surveillance in Somalia: results from nineteen cross-sectional surveys (2007-2016)

Authors:

Estefania Custodio (estefania.custodio@ec.europa.eu)
Rocio Martin-Cañavate (rociomartincanavate@gmail.com)
Federica Di Marcantonio (Federica.di-marcantonio@jrc.ec.europa.eu)
Daniel Molla (daniel.molla@fao.org)
Abukar Yusuf (abukar.yusuf@fao.org)
Francois Kayitakire (francois.kayikatire@ec.europa.eu)

Version: 4 Date: 23 Jan 2018

Author’s response to reviews:

Dear Editor:

We have replaced the Box 1 by a Table 1, and changed all the rest of the table names accordingly.

We hope this time formatting requests are fulfilled.

Please let us know if otherwise.

Looking forward to hearing from you.

With kind regards,

Estefania Custodio